1. The article by Yang et al. (Cell Death and Disease, 2018, 9:793) elucidates the mechanism by which Wnt signaling promotes TAM polarization in liver cancer. This should be considered in comparison with the work of Huang et al. discussed in this editorial.

Thank you for your comments. I have cited this article and included the discussion in the text.

2. It is essential to include a discussion of the innovations and limitations presented in Huang et al.’s study.

Thank you for your comments. I have added a sentence to emphasize the strength of this study in the text.